Author's response to reviews

Title: Intraoperative electrocortical stimulation of Brodman area 4: A 10-year analysis of 255 cases.

Authors:

Olaf Suess (olaf.suess@charite.de)
Silke Suess (silke.suess@charite.de)
Mario Brock (mario.brock@charite.de)
Theodoros Kombos (theodoros.kombos@charite.de)

Version: 3 Date: 2 July 2006

Dear Editors,

thank you for publishing the above mentioned manuscript. Please find enclosed the "final version" with the following corrections:
1) "acknowledgements" is removed as it is unnecessary
2) "header" is removed
3) Figures 6-8 A-E were renamed as Figure 6, 7I+II and 8I+II, changes in the text were made according to the renaming

Once again thank you for accepting our paper.
With kindest regards,
Dr. O.Suss